Author’s response to reviews

Title: Identification of a novel nonsense mutation in SH2D1A gene from a patient with X-linked lymphoproliferative syndrom type 1: a case report

Authors:

Xiaodong Lyu (xiaodonglv2007@sina.com)
Zhen Guo (gzheng2008@163.com)
Yangwei Li (liyangweimry@163.com)
Ruifeng Fan (13653866010@163.com)
Yongping Song (ypsong2017@163.com)

Version: 3 Date: 22 Feb 2018

Author’s response to reviews:

Editor Comments:

The authors have answered only some of the reviewers comments particularly regarding changing the main conclusion towards the novel mutation (and not methodology) and grammar/spelling mistakes.

Please have the text edited by a professional language editing service or a native English speaking colleague (see below) after introducing all the necessary changes.

Thorough language revision is mandatory in order to make the manuscript suitable for acceptance.

Reviewer reports:

Please revised the whole manuscript and not only grammar/spelling mistakes mentioned in the attachment.
Answer: We have received assistance from a professional editing agency. The whole manuscript has been revised by a native English speaker. A certification of English editing had been attached. We hope that it is now suitable for publication.